Author's response to reviews

Title: Mobility of primary health workers in China

Authors:

Qingyue Q Meng Prof (qmeng@sdu.edu.cn)
Jing J Yuan Dr (yuanjing@mail.sdu.edu.cn)
Limei L Jing Ms (limei100409@mail.sdu.edu.cn)
Junhua J Zhang Dr (jzhang70@yahoo.com)

Version: 3 Date: 8 January 2009

Author's response to reviews:

In review reports from the two reviewers, there are no specific comments and suggestions for revising the manuscript. Positive support for publishing this manuscript from the reviewers is much appreciated. For making the manuscript more readable, the authors did revisions by polishing the language. No major changes on content were made.